Author's response to reviews

Title: Social Action or Critical Analysis? A concept analysis of critical health literacy

Authors:

Susie Sykes (susiesykes@mac.com)
Jane Wills (willsj@lsbu.ac.uk)
Gillian Rowlands (rowlang2@lsbu.ac.uk)
Keith Popple (popplekj@lsbu.ac.uk)

Version: 2 Date: 21 July 2012

Author's response to reviews:

a sentence has been added to the methodology section confirming that the research complies with the principles of the Helsinki declaration and that it was submitted to and approved by London South Bank University’s Research Ethics Committee.